Author’s response to reviews

Title: Mesenteric cysts and mesenteric venous thrombosis led to intestinal necrosis in pregnancy and managed with laparotomy. Case report and review of the literature

Authors:

Aris Giannos (arisgiannos@yahoo.gr)
Sofoklis Stavrou (sfstavrou@yahoo.com)
Nikolaos Goumalatsos (ngoum@yahoo.gr)
George Fragkoulidis (fragkgeo@yahoo.gr)
Chra Eleni (eleni.chra@yahoo.com)
Dimitrios Argiropoulos (argidimi1980@yahoo.it)
Peter Drakakis (pdrakakis@hotmail.com)
Dimitrios Loutradis (LOUTRADI@OTENET.GR)

Version: 1 Date: 16 Mar 2017

Author’s response to reviews:

JMCR-D-16-00683 manuscript

Point-by- point response to reviewer 2:

1. Thank you for your comment, we made the relevant amendments in the abstract.
2. Thank you for your comment, we checked the punctuation throughout the whole manuscript.
3. Thank you for your comment, we deleted all the repetitions of our references.
4. Thank you for your comment, we made the amendments in our case presentation section.
5. Thank you for your comment, we added a relevant text on page 3(line 31-36) to 4(line1) (intense black color): The obstetrical and .......were revealed and the word: by radiologists on page 4, line 7.
6. Thank you for your comment, we added a relevant text on page 4, line 8-10 (intense black color): Due to increased BMI of the patient, the history of previous laparotomy and the lack of available qualified specialists for laparoscopy at that time.

7. Thank you for your comment, we added a relevant text on page 4, line 25-26 (intense black color): She was mobilized very early wearing antithrombotic socks continuously.

8. Thank you for your comment, we made the relevant amendments, so the language was strongly revised. The amendments are in red color throughout the whole manuscript.

9. Thank you for your comment, the medical terms were corrected (red color) throughout the whole manuscript.

10. Thank you for your comment, we shortened the discussion section as you said (from 1258 words to 974).

11. Thank you for your comment, we made the amendments in conclusion section, which was restructured according to your instructions.